Author's response to reviews

Title: Geographical differences in whooping cough in Catalonia, Spain, from 1990 to 2010.

Authors:

Inma Crespo (inma.crespof@gencat.cat)
Núria Soldevila (nuriasolde_@hotmail.com)
Pilar Muñoz (pilar.munyoz@upc.edu)
Pere Godoy (pere.godoy@gencat.cat)
Gloria Carmona (gloria.carmona@gencat.cat)
Angela Domínguez (angela.domiguez@ub.edu)

Version: 3 Date: 25 July 2013

Author's response to reviews: see over
Dear Editor,

I send you a description of the changes I have made at the manuscript.

1. I have changed the affiliation of the third author Pilar Muñoz, the new one is: Departament d'Estadística i Inv. Operativa, Edifici C5 (campus nord), C/ Jordi Girona, 1-3; 08034 Barcelona, Spain.

2. I have included a description of where we have obtained the data used in this article, which is freely available.

3. I have made some changes in author’s contribution.

Thank you,

Kind regards,

25th July, 2013

Inma Crespo, MSc (corresponding author)

Inma.crespof@gencat.cat

CIBER Epidemiology and Public Health;

Public Health Department. University of Barcelona (UB)

08036, Barcelona, Spain.

Tel: +34934024566